Author's response to reviews

Title: Health related quality of life measured by SF-36: a population-based study in Shanghai, China

Authors:

rui wang (fengxinzimail@hotmail.com)
chenguwu (wucheng_wu@hotmail.com)
yanfangzhao (zhyf115@hotmail.com)
xiaoyanyan (ghost_tian@hotmail.com)
xiuqiangma (mxqiang1823@hotmail.com)
meijingwu (wenny2700@hotmail.com)
wenbiniu (Wilbur.Liu@astrazeneca.com)
zhenggu (guzheng2003@hotmail.com)
juen Zhao (June.Zhao@astrazeneca.com)
jiahe (hejia63@yahoo.com)

Version: 6 Date: 8 August 2008

Author's response to reviews:

Dear Drs. Iratxe Puebla and Rikki Graham:

We have revised the manuscript according to the comments. Attached please find our responses to the comments.

Best wishes,
Rui Wang

Revisions

1. Abbreviations- please format the abbreviations as a paragraph rather than as a list.

Response:
The abbreviations were defined in the text where first used.

2. Tables must be created using a table’s tool. Tables must be divided into rows and columns and be orientated in a portrait page layout, without visible vertical lines.

Response:
We have revised that, all tables on a landscape page were reformatted onto a portrait page.